Author’s response to reviews

Title: Role of enteral nutrition in nonthyroidal illness syndrome: a retrospective observational study

Authors:
Ranran Li (doublerans@hotmail.com)
Jianan Ren (jiananr@gmail.com)
Qin Wu (qinwu0221@gmail.com)
Gefei Wang (wgfwang@gmail.com)
Xiuwen Wu (lygwxxw@163.com)
Jun Chen (yccjun@163.com)
Guanwei Li (li.guanwei@foxmail.com)
Zhiwu Hong (zhwhong1986@163.com)
Huajian Ren (rhj1288@163.com)
Yunzhao Zhao (yzzhaomd@gmail.com)
Jieshou Li (jieshou_li@163.com)

Version: 3 Date: 21 Oct 2015

Author’s response to reviews:

Editor’s suggestions

1. "Individual contribution"

Please rename this section to "Authors’ Contributions“ and move it from the title page to after section "Competing interests". Please also note that currently the set of initials do not match the authors list - author YL is not mentioned, and the initials GG and NA do not seem to correspond to any name.

Answers: We have moved the "Authors’ Contributions" to the right place. Also, we have checked the author list as well as the "Authors' Contributions".

2. Copyediting.
We recommend that you ask a native English speaking colleague to help you copyedit the paper.

Answers: we have invited Dr. Kerya Martinez Dunn to help us copyedit the paper as a native English speaking colleague.

3. Emails missing.

Please provide the email addresses of all authors on the title page.

Answers: email address for all authors are listed on the title page.


Please add a section "Conclusions" after section "Discussion" in the main text. You may for example consider using the contents of the last paragraph of section "Discussion" for this purpose.

Answers: we have added the “Conclusions” section in our manuscript.

5. Acknowledgements

You may wish to consider adding a section ‘Acknowledgements’ before the reference list. There you can acknowledge anyone who made substantial contributions towards the article, but who does not meet the criteria for authorship, and the source(s) of funding for each author.

Answers: Acknowledgements section has been added in our manuscript.

6. Section titles

Please change section titles "Introduction" to "Background", "Materials and Methods" to "Methods", and "A competing interest statement" to "Competing interests".

Answerers: we have changed the section titles.